CME MONOGRAPH

VISIT HTTPS://TINYURL.COM/SAVINGSIGHTNAMD FOR ONLINE TESTING AND INSTANT CME CERTIFICATE.



THE LATEST IN CLINICAL TRIALS AND BEST PRACTICES

Original Release: July 1, 2020 • Expiration: July 31, 2021



FACULTY

Nancy M. Holekamp, MD (Chair)



David Eichenbaum, MD



Arshad M. Khanani, MD, MA



Michael Singer, MD

This continuing medical education activity is provided by **New York Eye and Ear Infirmary of Mount Sinai**. This educational activity was developed and implemented in collaboration with **MedEdicus LLC**.





This continuing medical education activity is supported through an unrestricted educational grant from Allergan.



LEARNING METHOD AND MEDIUM

This educational activity consists of a supplement and ten (10) study questions. The participant should, in order, read the learning objectives contained at the beginning of this supplement, read the supplement, answer all questions in the post test, and complete the Activity Evaluation/Credit Request form. To receive credit for this activity, please follow the instructions provided on the post test and Activity Evaluation/Credit Request form. This educational activity should take a maximum of 1.5 hours to complete.

ACTIVITY DESCRIPTION

By 2030, it is estimated that 3.7 million people in the United States will have advanced age-related macular degeneration, including neovascular age-related macular degeneration (nAMD) and geographic atrophy. Current treatments for nAMD leave much to be desired in terms of efficacy and treatment burden. New and emerging treatments for nAMD use novel molecules, delivery modalities, and targets to achieve better treatment longevity and reduced treatment burden. These include a DARPin (designed ankyrin repeat protein) (abicipar pegol), a small anti-vascular endothelial growth factor (VEGF) monoclonal antibody fragment (brolucizumab), an implantable delivery system (ranibizumab port delivery), and a monoclonal antibody with multiple angiogenic targets (faricimab). Vision maintenance, change in bestcorrected visual acuity, and retinal fluid resolution are among the end points being explored vs traditional anti-VEGF agents in phase 2 and 3 clinical trials, with promising results. Developing retreatment plans for patients should balance fluid resolution and injection burden, considering a treat-and-extend approach. This activity captures the proceedings of a live roundtable discussion of expert retina specialists held during the Angiogenesis, Exudation, and Degeneration 2020 meeting. The desired results of this educational activity are for retina specialists and other ophthalmologists to evaluate emerging treatment strategies for nAMD in the context of the current standard of care.

TARGET AUDIENCE

This educational activity is intended for retina specialists and other ophthalmologists caring for patients with nAMD.

LEARNING OBJECTIVES

Upon completion of this activity, participants will be better able to:

- · Describe the mechanism of extended therapeutic effect for investigational treatments for nAMD
- Discuss clinical trial data for approved and emerging treatments for nAMD
- Develop retreatment plans for patients with nAMD guided by recent clinical trial results

ACCREDITATION STATEMENT

The New York Eye and Ear Infirmary of Mount Sinai is accredited by the Accreditation Council for Continuing Medical Education (ACCME) to provide continuing medical education for physicians. This educational activity was ACCME developed and implemented in collaboration with MedEdicus LLC.

AMA CREDIT DESIGNATION STATEMENT

The New York Eye and Ear Infirmary of Mount Sinai designates this enduring material for a maximum of 1.5 AMA PRA Category 1 Credits™. Physicians should claim only the credit commensurate with the extent of their participation in the activity.

GRANTOR STATEMENT

This continuing medical education activity is supported through an unrestricted educational grant from Allergan

DISCLOSURE POLICY STATEMENT

It is the policy of New York Eye and Ear Infirmary of Mount Sinai that the faculty and anyone in a position to control activity content disclose any real or apparent conflicts of interest relating to the topics of the educational activity in which they are participating. They are also required to disclose discussions of unlabeled/unapproved uses of drugs or devices during their presentations. New York Eye and Ear Infirmary of Mount Sinai is committed to providing its learners with quality CME activities and related materials that promote improvements in healthcare and not the proprietary interests of a commercial interest and, thus, has established policies and procedures in place that identify and resolve all conflicts of interest prior to the execution or release of its educational activities. Full disclosure of faculty/planners and their commercial relationships, if any, follows.

DISCLOSURES

David Eichenbaum, MD, had a financial agreement or affiliation during the past year with the following commercial interests in the form of Consultant/Advisory Board: Alimera Sciences; Allergan; Clearside Biomedical, Inc; D.O.R.C. Dutch Ophthalmic Research Center (International) B.V.; EyePoint Pharmaceuticals; Genentech, Inc; Gyroscope; Kodiak Sciences Inc; Notal Vision; Novartis Pharmaceuticals Corporation; Recens Medical; and Regeneron Pharmaceuticals, Inc; Contracted Research: Alimera Sciences; Allergan; Chengdu Kanghong Pharmaceutical Group Co Ltd; Clearside Biomedical, Inc; Genentech, Inc; Gyroscope; IVERIC bio; Kodiak Sciences Inc; Mylan NV; NGM Biopharmaceuticals; Novartis Pharmaceuticals Corporation; and Opthea; Honoraria from promotional, advertising or non-CME services received directly from commercial interests or their Agents (eg, Speakers Bureaus): Allergan; D.O.R.C. Dutch Ophthalmic Research Center (International) B.V.; Genentech, Inc; and Novartis Pharmaceuticals Corporation; Ownership Interest (Stock options, or other holdings, excluding diversified mutual funds): Clearside Biomedical, Inc; and Hemera Biosciences.

Nancy M. Holekamp, MD, had a financial agreement or affiliation during the past year with the following commercial interests in the form of Receipt of Intellectual Rights/ Patent Holder: Katalyst Healthcares & Life Sciences; Consultant/Advisory Board: Acucela Inc; Allergan; Clearside Biomedical, Inc; Gemini Therapeutics; Genentech, Inc; Gyroscope; Katalyst Healthcares & Life Sciences; Lineage Cell Therapeutics; Notal Vision; Novartis Pharmaceuticals Corporation; and Regeneron Pharmaceuticals, Inc; Contracted Research: Gemini Therapeutics; Genentech, Inc; Gyroscope; and Notal Vision; Honoraria from promotional, advertising or non-CME services received directly from commercial interests or their Agents (eg, Speakers Bureaus): Allergan; Genentech, Inc; Novartis Pharmaceuticals Corporation; Regeneron Pharmaceuticals, Inc; and Spark Therapeutics, Inc; Ownership Interest (Stock options, or other holdings, excluding diversified mutual funds): Katalyst Healthcares & Life Sciences.

Arshad M. Khanani, MD, MA, had a financial agreement or affiliation during the past year with the following commercial interests in the form of Consultant/Advisory Board: Alcon; Adverum; Allegro Ophthalmics, LLC; Allergan; Bausch & Lomb Incorporated; EyePoint Pharmaceuticals; Gemini Therapeutics; Genentech, Inc; GrayBug, Inc; Gyroscope; Kodiak Sciences Inc; Novartis Pharmaceuticals Corporation; Opthea; Oxurion NV; PolyPhotonix; Recens Medical; and Regenzbio Inc; Contracted Research: Adverum; Allegro Ophthalmics, LLC; Allergan; Gemini Therapeutics; Genentech, Inc; GrayBug, Inc; Gyroscope; IVERIC bio; Kodiak Sciences Inc; Novartis Pharmaceuticals Corporation; Opthea; Oxurion NV; Recens Medical; and Regenxbio Inc; Honoraria from promotional, advertising or non-CME services received directly from commercial interests or their Agents (eg, Speakers Bureaus): Allergan; and Novartis Pharmaceuticals Corporation.

Michael Singer, MD, had a financial agreement or affiliation during the past year with the following commercial interests in the form of Consultant/Advisory Board: Aerie Pharmaceuticals, Inc; Allergan; Ampio Pharmaceuticals Inc; Apellis Pharmaceuticals; Clearside Biomedical, Inc; Genentech, Inc; Ionis Pharmaceuticals, Inc; Kodiak Sciences Inc; Novartis Pharmaceuticals Corporation; Regeneron Pharmaceuticals, Inc; Santen Inc; and Spark Therapeutics, Inc; Contracted Research: Aerie Pharmaceuticals, Inc; Allergan; Ampio Pharmaceuticals Inc; Apellis Pharmaceuticals; Clearside Biomedical, Inc; Genentech, Inc; Ionis Pharmaceuticals, Inc; Kodiak Sciences Inc; Novartis Pharmaceuticals Corporation; Regeneron Pharmaceuticals, Inc; Santen Inc; and Spark Therapeutics, Inc; Honoraria from promotional, advertising or non-CME services received directly from commercial interests or their Agents (eg, Speakers Bureaus): Aerie Pharmaceuticals, Inc; Allergan; Ampio Pharmaceuticals Inc; Apellis Pharmaceuticals; Clearside Biomedical, Inc; Genentech, Inc; Ionis Pharmaceuticals, Inc; Kodiak Sciences Inc; Novartis Pharmaceuticals Corporation; Regeneron Pharmaceuticals, Inc: Santen Inc: and Spark Therapeutics, Inc.

NEW YORK EYE AND EAR INFIRMARY OF MOUNT SINAI PEER REVIEW DISCLOSURE

Gennady Landa, MD, has no relevant commercial relationships to disclose.

EDITORIAL SUPPORT DISCLOSURES

Erika Langsfeld, PhD; Cynthia Tornallyay, RD, MBA, CHCP; Kimberly Corbin, CHCP; Barbara Aubel; and Michelle Ong have no relevant commercial relationships to disclose.

Medical Writer: Tony Realini, MD, had a financial agreement or affiliation during the past year with the following commercial interests in the form of Consultant/Advisory Board: Aerie Pharmaceuticals, Inc; iSTAR; New World Medical, Inc; and Notal Vision.

DISCLOSURE ATTESTATION

The contributing physicians listed above have attested to the following: 1) that the relationships/affiliations noted will not bias or otherwise influence their involvement in this activity;

- 2) that practice recommendations given relevant to the companies with whom they have relationships/affiliations will be supported by the best available evidence or,
- absent evidence, will be consistent with generally accepted medical practice; and 3) that all reasonable clinical alternatives will be discussed when making practice recommendations

OFF-LABEL DISCUSSION

This CME activity includes discussion of unlabeled and/or investigative uses of drugs. Please refer to the official prescribing information for each drug discussed in this activity for FDA-approved dosing, indications, and warnings.

NEW YORK EYE AND EAR INFIRMARY OF MOUNT SINAI

PRIVACY & CONFIDENTIALITY POLICIES https://www.nyee.edu/education/cme

CME PROVIDER CONTACT INFORMATION For questions about this activity, call 212-870-8125.

TO OBTAIN AMA PRA CATEGORY 1 CREDIT™

To obtain AMA PRA Category 1 Credit™ for this activity, read the material in its entirety and consult referenced sources as necessary. Please take this post test and evaluation online by going to https://tinyurl.com/savingsightnAMD. Upon passing, you will receive your certificate immediately. You must score 70% or higher to receive credit for this activity, and may take the test up to 2 times. Upon registering and successfully completing the post test, your certificate will be made available online and you can print it or file it.

DISCLAIMER

The views and opinions expressed in this educational activity are those of the faculty and do not necessarily represent the views of New York Eye and Ear Infirmary of Mount Sinai, MedEdicus LLC, Allergan, or Retina.

This CME activity is copyrighted to MedEdicus LLC ©2020. All rights reserved. 213





PROGRAM CHAIR AND MODERATOR

Nancy M. Holekamp, MD

Professor of Clinical Ophthalmology Washington University School of Medicine Director, Retina Services Pepose Vision Institute St Louis, Missouri

FACULTY

David Eichenbaum, MD

Collaborative Associate Professor of Ophthalmology Morsani College of Medicine University of South Florida Partner and Director of Research Retina Vitreous Associates of Florida Tampa, Florida

Arshad M. Khanani, MD, MA

Managing Partner Director of Clinical Research Director of Fellowship Sierra Eye Associates Clinical Associate Professor University of Nevada, Reno School of Medicine Reno, Nevada

Michael Singer, MD

Clinical Professor of Ophthalmology University of Texas Health Science Center Director of Clinical Research Medical Center Ophthalmology Associates San Antonio, Texas

CME REVIEWER FOR NEW YORK EYE AND EAR INFIRMARY OF MOUNT SINAI

Gennady Landa, MD

Associate Professor of Ophthalmology Icahn School of Medicine at Mount Sinai Director of Retina Service Associate Director of Vitreoretinal Fellowship Medical Director of Tribeca Office Vitreoretinal Specialist and Attending Surgeon Department of Ophthalmology New York Eye and Ear Infirmary of Mount Sinai New York, New York

THERAPEUTIC LANDSCAPE FOR nAMD

Age-related macular degeneration (AMD) is the leading cause of blindness in developed regions of the world, and accounts for nearly 9% of all blindness worldwide.¹ In 2020, approximately 196 million people are affected with AMD worldwide; this number is projected to grow to 288 million by 2040.¹ The disease begins as dry—or nonneovascular—AMD, and, in some eyes, progresses to wet—or neovascular—AMD (nAMD). Vision loss can occur in eyes with advanced dry AMD (primarily from geographic atrophy involving the fovea), but most AMD-related vision loss arises from the development of choroidal neovascularization (CNV) associated with nAMD.² nAMD accounts for only 10% to 15% of all AMD, but is responsible for more than 80% of all AMD-related vision loss.³

Three inhibitors of vascular endothelial growth factor (VEGF)—ranibizumab, aflibercept, and brolucizumab—are approved by the US Food and Drug Administration (FDA) for the treatment of nAMD, and a fourth drug—bevacizumab—is often used off-label because of its cost advantage over the 3 indicated products. The efficacy and safety profiles of these drugs have been established in well-designed and appropriately powered phase 3 trials, and the safety of brolucizumab continues to be evaluated as real-world use expands.⁴ They are associated with significant treatment burden, however, requiring regular repeated injections every 1 to 3 months according to their labels. In the real world, these drugs are associated with a high injection frequency to maintain vision.⁵

Ranibizumab was compared with sham injection in MARINA (Minimally Classic/Occult Trial of the Anti-VEGF Antibody Ranibizumab in the Treatment of Neovascular AMD) and with photodynamic therapy (PDT) with verteporfin in the ANCHOR (Anti-VEGF Antibody for the Treatment of Predominantly Classic Choroidal Neovascularization in AMD) study.^{6,7} In MARINA, ranibizumab 0.3 mg (n = 238) and 0.5 mg (n = 240) administered by monthly intravitreal injection maintained best-corrected visual acuity (BCVA) better than did sham therapy (n = 238), with 92%, 90%, and 53% of eyes, respectively, losing < 15 letters of BCVA at 24 months (P < .001 for each ranibizumab dose vs sham).⁶ In ANCHOR, 90%, 90%, and 66% of eyes receiving ranibizumab 0.3 mg (n = 140), ranibizumab 0.5 mg (n = 140), or PDT (n = 143), respectively, attained the same primary end point (P < .0001 for each ranibizumab dose vs PDT).7 Low rates of endophthalmitis-less than 1%-were seen in these trials and were attributed to the injection procedure rather than the drug. Other uncommon serious ocular adverse events included low-grade inflammation, retinal detachments, and vitreous hemorrhages.^{6,7} Ranibizumab was also compared with off-label bevacizumab in CATT (Comparison of Age-Related Macular Degeneration Treatments Trials), which demonstrated comparable BCVA outcomes at 2 years.⁸ superior BCVA outcomes with monthly vs as-needed injections,⁸ superior fluid resolution with ranibizumab vs bevacizumab,⁹ and comparable ocular safety of the 2 agents.8

Aflibercept was compared with ranibizumab in the VIEW 1 and VIEW 2 (Vascular Endothelial Growth Factor Trap-Eve: Investigation of Efficacy and Safety in Wet Age-Related Macular Degeneration) studies.¹⁰ In VIEW 1, < 15 letters of BCVA at 12 months (the primary end point of the VIEW 1 and VIEW 2 studies) was lost by 95% of 301 patients receiving aflibercept 0.5 mg every 4 weeks, by 95% of 304 patients receiving aflibercept 2 mg every 4 weeks, by 94% of 301 patients receiving affibercept 2 mg every 8 weeks. and by 94% of 304 patients receiving ranibizumab 0.5 mg every 4 weeks. In VIEW 2, the primary end point was attained by 95% of 296 patients receiving aflibercept 0.5 mg every 4 weeks, by 95% of 309 patients receiving aflibercept 2 mg every 4 weeks, by 95% of 306 patients receiving aflibercept 2 mg every 8 weeks, and by 95% of 291 patients receiving ranibizumab 0.5 mg every 4 weeks. In both studies, all doses of aflibercept were noninferior to ranibizumab for this primary end point. Aflibercept 2 mg every 4 weeks more effectively dried the macula, and eyes with early persistent fluid had better BCVA outcomes with every-4-week dosing than with every-8-week dosing.¹¹ The safety profiles of all 4 treatments were similar; serious ocular adverse events were uncommon and included endophthalmitis, reduced visual acuity (VA), and retinal hemorrhage.¹⁰ Arterial thromboembolic events were rare and had a comparable incidence among the groups.¹⁰



Brolucizumab, an antibody fragment, is the most recently approved drug for nAMD. This newly approved agent was designed to confer a reduced treatment frequency compared with the other anti-VEGF agents. It is thought to achieve a prolonged duration of activity given its small size and higher molar dose compared with aflibercept, bevacizumab, or ranibizumab. The phase 3 HAWK and HARRIER trials compared brolucizumab 3 mg every 12 weeks (n = 358; HAWK only), brolucizumab 6 mg every 12 weeks (n = 360 in HAWK; n = 370 in HARRIER), and aflibercept 2 mg every 8 weeks, the FDA-approved dose at the time of the study (n = 360 in HAWK; n = 369 in HARRIER). As part of a flexible trial model, the brolucizumab dosing interval was reduced to every 8 weeks if disease activity was observed at week 16 and at each assessment thereafter in the every-12-week arm of the study.¹² The primary efficacy outcome was change from baseline in mean BCVA at 12 months. In HAWK, mean BCVA changes from baseline were +6.1 letters, +6.6 letters, and +6.8 letters with brolucizumab 3 mg, brolucizumab 6 mg, and aflibercept, respectively, with both brolucizumab groups being noninferior to the aflibercept group. In HARRIER, the mean BCVA change from baseline with brolucizumab 6 mg (+6.9 letters) was also noninferior to that seen with aflibercept (+7.6 letters). Overall, 51% to 56% of eyes receiving brolucizumab 6 mg every 12 weeks were maintained with every-12-week dosing through 12 months in these 2 trials. In both studies, fewer eyes receiving brolucizumab 6 mg had any subretinal and/or intraretinal fluid on optical coherence tomography (OCT) images at months 4 and 12 compared with eyes receiving aflibercept (relative risk reductions of approximately 30%-50%; P < .001 across studies and time points), and mean central subfield thickness was lower at month 12 in eyes receiving brolucizumab than in eyes receiving affibercept (P < 0.001). A recent 24-month analysis from these studies confirmed the durability of these findings through 2 years of treatment, although by the end of month 24, only 45% of 360 HAWK patients and 39% of 370 HARRIER patients on brolucizumab 6 mg were still maintained on every-12-week retreatment (Figure 1).^{13,14} The most common adverse events associated with brolucizumab are blurred vision (10%), cataract (7%), conjunctival hemorrhage (6%), vitreous floaters (5%), eye pain (5%), and intraocular inflammation (4%).¹³ Postmarketing cases of retinal vasculitis and retinal vascular occlusion have occurred, prompting the manufacturer to launch an adverse event reporting site to collect and share pertinent data.4

CURRENT nAMD TREATMENT BURDEN IS UNSUSTAINABLE

The clinical trials described previously,^{6-12,15} coupled with many others,¹⁶⁻²¹ convincingly demonstrate that consistent anti-VEGF dosing at a regular interval—eg, every 1, 2, or 3 months—produces significant VA gains in eyes with nAMD; these gains are greater than those achieved with irregular administration of as-needed anti-VEGF therapy when its effect begins to wane.^{8,22} More injections lead to greater VA gains (**Figure 2**).²³

The need for frequent ongoing injections imposes a tremendous treatment burden on patients and health care providers alike. A consequence of this burden is that patients do not receive as many injections as they perhaps should. Lower injection rates lead to smaller VA gains and can lead to loss of initial gains when injection frequencies decrease over time. This was demonstrated in the real-world European **AURA** study, in which national cohorts with lower injection rates experienced smaller VA gains, and eyes receiving fewer than 6 injections per year lost VA over time.⁵ A similar result was seen in the **HORIZON** extension of the MARINA, ANCHOR, and **FOCUS** studies, in which injection rates and vision gains fell off rapidly when patients exited the rigid protocol-directed treatment phase and entered the real-world observation



Figure 1. Best-corrected visual acuity changes from baseline in the HAWK and HARRIER phase 3 clinical trials of brolucizumab^{13,14}

^a Noninferiority margin = 4 letters

Abbreviations: BCVA, best-corrected visual acuity; ETDRS, Early Treatment Diabetic Retinopathy Study; LS, least squares; SE, standard error. Permission request submitted.



Figure 2. Relationship between the number of anti–vascular endothelial growth factor injections and visual acuity gains across randomized trials²³

Abbreviations: ETDRS, Early Treatment Diabetic Retinopathy Study; inj, injection; LG, letters gained.

Reprinted from *American Journal of Ophthalmology*, 157, Holekamp NM, Liu Y, Yeh WS, et al, Clinical utilization of anti-VEGF agents and disease monitoring in neovascular age-related macular degeneration, 825-833.e1, Copyright 2014, with permission from Elsevier.

with investigator-driven injection frequency phase.²⁴ More broadly, a comparison of first-year injection rates in randomized clinical trials compared with real-world studies **(Figure 3)** demonstrates the extent to which anti-VEGF therapy is underdosed in clinical practice, at the cost of smaller visual gains.^{6,7/0,16-21,25}





Figure 3. Disconnect between injection rates in clinical trials and clinical practice, with lesser efficacy with lower injection rates in real-world practice $^{6.710.16-21.25}$

Abbreviations: BCVA, best-corrected visual acuity; EMR, electronic medical record; ETDRS, Early Treatment Diabetic Retinopathy Study; q4w, 4-week dosing interval; q8w, 8-week dosing interval; RCT, randomized controlled trial. * Ranibizumab monthly and aflibercept bimonthly dosing unless stated

REDUCING nAMD TREATMENT BURDEN: HOME OCT AND TREAT-AND-EXTEND ANTI-VEGF DOSING STRATEGY

The current approach to nAMD monitoring and therapy is costly and, as demonstrated previously, suboptimally effective in the real-world setting. There is a significant unmet need for more efficient monitoring and treatment strategies that reduce the burden of nAMD management.

Improved home monitoring for patients with AMD has the potential to both reduce the frequency of office visits for disease activity assessment and detect conversion to nAMD earlier than by traditional means. Early detection is crucial for preserving VA because baseline VA is a powerful predictor of final VA in eyes with new-onset nAMD receiving anti-VEGF therapy.^{26,27} A patient-operated home OCT device (Notal Vision) is in development; it has demonstrated a sensitivity of 91.5% and a specificity of 97% vs conventional OCT operated by trained personnel in detecting intraretinal and/or subretinal fluid in the macula of eyes with

new-onset nAMD.²⁸ In a study of 347 eyes of 196 patients with intermediate AMD, an average age of 77 years, and an average VA of 20/40 (with 20% having VA of 20/100 or worse), 90% of patients were able to successfully complete the self-scan in at least 1 eye, demonstrating the usability of the technology.

Treatment burden can also be decreased by reducing the frequency of injections. One way to accomplish this is to develop drugs with longer durations of action. The initial label dose frequency was monthly for ranibizumab, up to every 2 months for aflibercept, and up to every 3 months for brolucizumab, each after appropriate loading.^{13,29,30} At present, although the current labels for all 3 drugs include options for dosing every 3 months, the labels for ranibizumab and aflibercept—but not for brolucizumab—state that this dosing interval may be less effective than more frequent dosing.^{13,29,30}

Another option for reducing injection frequency is to recognize that response to anti-VEGF therapy is heterogeneous among patients, and that some patients can be dosed less frequently than indicated by the drug labels. The as-needed dosing strategy—in which eyes are retreated only upon worsening of VA and/or OCT findings—has been shown to be comparable to monthly dosing with frequent monitoring.^{8,16,31} Effective as-needed dosing is practically difficult because it necessitates regular—likely monthly—disease activity assessments to identify the need for retreatment and duplicate the results of the noninferiority **HARBOR** trial.

More recently, clinical practice has adopted the treat-and-extend (TAE) approach to anti-VEGF therapy. In this strategy, following the appropriate loading dose, intervals between disease assessment visits are progressively lengthened in 2- to 4-week increments to identify the maximum duration of effect of a given drug in a given patient. Upon recurrence of disease activity, the reassessment interval is typically reduced by 2 to 4 weeks.

Several studies have demonstrated the clinical value of the TAE approach to anti-VEGF dosing (summarized in **Table 1**).³²⁻³⁷ The **TREX-AMD** (Treat-and-Extend Protocol in Patients With Wet Age-Related Macular Degeneration) study compared monthly ranibizumab with TAE ranibizumab over 24 months in eyes with nAMD.³² Eyes in the TAE group received at least 3 monthly loading

Table 1. Efficacy Summary of Key Studies Evaluating Treat-and-Extend Dosing Strategies for

 Neovascular Age-Related Macular Degeneration

| Study | Drug | Comparator | Duration, months | Findings | |
|------------------------|--|--|---------------------|--|--|
| TREX-AMD ³² | Ranibizumab TAE | Ranibizumab monthly | 24 | BCVA similar Fewer injections with TAE | |
| TREND ³³ | Ranibizumab TAE | Ranibizumab monthly | 12 | BCVA similar (noninferior) OCT images similar Fewer injections with TAE | |
| CANTREAT ³⁴ | Ranibizumab TAE | Ranibizumab monthly | 12 | BCVA similar Fewer injections with TAE | |
| LUCAS ³⁵ | Ranibizumab TAE | Bevacizumab TAE | 24 | BCVA similar OCT images similar Fewer injections with ranibizumab | |
| ATLAS ³⁶ | Aflibercept TAE | None | 24 | BCVA improved from baseline in year 1 but less so in year 2 OCT images similar in years 1 and 2 Fewer injections in year 2 | |
| ALTAIR ³⁷ | Aflibercept TAE 2-week extension | Aflibercept TAE 4-week extension | 24 | BCVA similar OCT images similar Injection rates similar | |

Abbreviations: BCVA, best-corrected visual acuity; OCT, optical coherence tomography; TAE, treat and extend.



doses until resolution of disease activity by both examination and OCT criteria, after which the interdose interval was extended in 2-week increments to a maximum of 12 weeks. Among the 40 TAE eyes and 20 monthly eyes, mean BCVA gains at 24 months were similar (8.7 vs 10.5 letters, respectively; P = .6), whereas the mean number of injections was significantly lower for the TAE eyes (18.6 vs 25.5, respectively; P < .001). The mean maximum extension interval was 8.5 weeks, and 17% of patients were successfully extended to 11 to 12 weeks between retreatments. Ocular adverse events (eg, worsening cataract, epiretinal membrane, and progressive macular atrophy) were seen in 3% to 13% of TAE eyes, but in no monthly eyes.

The **TREND** (Treat and Extend) study also compared monthly ranibizumab (n = 327) with TAE ranibizumab (n = 323) for nAMD, but was a much larger study, with 650 treatment-naïve eyes.³³ Eyes in the TAE group began extending after the second monthly injection at month 1, in 2-week increments, to a maximum of 12 weeks, with 2-week reductions upon recurrence of disease activity. At 12 months, BCVA gains were 7.9 letters in the monthly group and 6.6 letters in the TAE group; this difference met the study's prespecified margin for noninferiority. Likewise, there was no significant difference in OCT central subfield thickness at 12 months between the monthly and TAE groups (-173 vs -169 µm, respectively; P = .75). The TAE group had fewer injections (8.7 vs 11.1, respectively) and fewer postbaseline visits (8.9 vs 11.2, respectively) than the monthly group. Common adverse events included elevated intraocular pressure and subconjunctival hemorrhage and were of equal frequency between groups.

CANTREAT (Canadian Treat-and-Extend Analysis Trial With Ranibizumab) was a comparison of monthly ranibizumab (n = 258) with TAE ranibizumab (n = 268) conducted in Canada.³⁴ This study included 526 eyes, and, as in other studies, treatment was extended in the TAE group after cessation or stability of nAMD disease activity, with shortening of intervisit intervals upon recurrence. At 12 months, noninferiority of TAE to monthly therapy was demonstrated, with mean BCVA gains of 8.4 and 6.0 letters in the TAE and monthly groups, respectively. This was accomplished with fewer injections in the TAE group (9.4 vs 11.8; P < .001). The nature of adverse events was not described, but the frequency was similar between groups.

LUCAS compared ranibizumab and bevacizumab when both were given using the TAE strategy.³⁵ Monthly loading doses were given to achieve disease inactivity, after which the dosing interval was extended in 2-week increments to a maximum of 12 weeks. After 2 years of therapy, visual gains (6.6 vs 7.4 letters, respectively; P = .63) and decrease in central retinal thickness (122 vs 113 μ m, respectively; P = .48) were similar in the patients receiving ranibizumab (n = 172) or bevacizumab (n = 167), whereas the number of injections was lower in the ranibizumab group (16.0 vs 18.2, respectively; P ≤ .001). At 2 years, 28% of ranibizumab eyes vs 45% of bevacizumab eyes had residual fluid on OCT. In this study, 2-week reductions in between-visit intervals were effective upon recurrence of disease activity, except for those eyes already maximally extended to 12 weeks, suggesting that more drastic reductions may be warranted in eyes with recurrence at 12-week extension. Although the numbers of each event were small (eg, 3 pseudoendophthalmitis, 3 macular hemorrhage, and 2 BCVA loss > 30 letters), all 11 serious adverse events reported in this study occurred in the bevacizumab group.

ATLAS (Aflibercept Treat and Extend for Less Frequent Administration Study) was a nonrandomized, uncontrolled, prospective interventional case series in which 40 eyes with nAMD were treated with aflibercept using the TAE dosing strategy for 2 years.³⁶ All eyes received monthly injections until prespecified criteria for disease inactivity were met, after which treatments were extended in 2-week intervals up to a maximum of 16 weeks, with 2-week reductions upon disease recurrence. The mean number of injections in years 1 and 2 was 8.0 and 6.5, respectively, which produced BCVA improvements of 7.2 letters (P < .001 for change from baseline) and 2.4 letters (P = .27), respectively. The reduced VA gains in year 2 were not related to loss of exudative control because mean change in central foveal thickness at 1 and 2 years was -209 and -211 µm, respectively. Instead, the investigators postulated that a small number of outlier patients with significant VA reductions during year 2 of the study may have reduced the overall mean BCVA change in a manner that was not representative of outcomes in most patients in the study. Ocular adverse events were limited to a single case of culture-positive endophthalmitis.

The **ALTAIR** study was a comparison of 2 TAE strategies of aflibercept; patients were randomly assigned to either a 2-week (n = 124) or 4-week (n = 123) extension after 3 monthly loading doses.³⁷ Both TAE strategies produced similar BCVA gains at 1 year (mean of 9.0 letters vs 8.4 letters, respectively) and 2 years (mean of 7.6 letters vs 6.1 letters, respectively). Likewise, changes in central retinal thickness were similar at 1 year (-134 μ m vs -126 μ m, respectively) and 2 years (-131 μ m vs -125 μ m, respectively). Both groups required a mean of 10.4 injections at 2 years, and approximately 60% were successfully extended to 12 weeks and 40% to 16 weeks. The nature and rates of ocular events (eg, cataract, conjunctival hemorrhage, and dry eye) were similar in both groups as well.

EMERGING THERAPIES FOR nAMD

The current array of anti-VEGF drugs—ranibizumab, bevacizumab, aflibercept, and, most recently, brolucizumab—has collectively driven a paradigm shift in the management of nAMD and other retinal vascular diseases. These drugs provide stability of VA in most treated patients and the potential for improved VA in a subset of these patients. Some patients, however, manifest incomplete responses to these agents or require ongoing monthly retreatment. There remains an unmet need for a broader armamentarium of therapies to prevent vision loss in eyes with nAMD while also reducing the treatment burden imposed by current therapies. Numerous novel molecules and drug delivery systems in various stages of clinical development seek to fulfill this unmet need.

Abicipar Pegol

Abicipar pegol (abicipar) is a DARPin (designed ankyrin repeat protein). These designer molecules contain an ankyrin repeat domain that can be engineered to bind to any desired target protein with high specificity and binding affinity,³⁸ and, in many cases, are heat stable to temperatures in excess of 80°C.³⁹ In the case of abicipar, its molecule target is all isoforms of VEGF-A (Figure 4A).^{40,41} Like all DARPins, abicipar is a small molecule (34 kDa), which allows for higher dosing on a molar basis (compared with full-sized anti-VEGF antibodies) and may enhance tissue penetration.⁴⁰ After a single intravitreal injection of abicipar 0.4 mg in an early phase 1/2 study in human eyes with diabetic macular edema, the ocular half-life of abicipar was ≥ 13 days, and the median aqueous humor concentration at 3 months remained several orders of magnitude above the half-maximal inhibitory concentration-the concentration needed for ongoing therapeutic anti-VEGF activity (Figure 4B).42

Abicipar is in late-stage clinical development for nAMD and is currently under review by both the FDA and the European Medicines Agency.⁴³ Two double-masked, randomized, phase 3 clinical trials—**CEDAR** and **SEQUOIA**—were recently completed.⁴⁴ Both studies compared abicipar 2 mg dosed at fixed intervals of either every 8 weeks or every 12 weeks with ranibizumab 0.5 mg



Figure 4. (A) Structural representation of a designed ankyrin repeat protein molecule. (B) Abicipar pegol concentration (nM) in the aqueous humor of patients with diabetic macular edema over 12 weeks after a single injection.⁴⁰⁻⁴² Abbreviations: ivt, intravitreal; PEG, polyethylene glycol.

Figure 4A reprinted by permission from Springer Nature Customer Service Centre GmbH: Springer Nature *Eye (London)*, Abicipar pegol: the nonmonoclonal antibody anti-VEGF, Sharma A, Kumar N, Kuppermann BD, Bandello F, 2020.

Figure 4B reprinted from American Journal of Ophthalmology, 155, Campochiaro PA, Channa R, Berger BB, et al, Treatment of diabetic macular edema with a designed ankyrin repeat protein that binds vascular endothelial growth factor: a phase I/II study, 697-704, Copyright 2013, with permission from Elsevier.

dosed monthly. The primary outcome was the proportion of patients with stable BCVA (losing < 15 letters from baseline) at month 12. In a pooled analysis of data from these 2 trials encompassing 1638 subjects that finished the full 104 weeks of the study, this outcome was achieved by 96% of eyes receiving abicipar every 8 weeks, by 94% of eyes receiving abicipar every 12 weeks, and by 97% of eyes receiving ranibizumab at month 12; by month 24, these numbers remained stable at 93%, 90%, and 94%, respectively (Table 2).44,45 Likewise, the mean changes from baseline in BCVA were 8.9, 7.4, and 9.5 letters, respectively, at month 12, and 7.8, 6.1, and 8.5 letters, respectively, at month 24. Central retinal subfield thickness on OCT was similarly improved in all 3 groups, with reductions of 143, 148, and 145 µm, respectively, at month 12. These comparable outcomes were achieved with vastly different treatment burdens: a total of 10 quarterly injections of abicipar vs 25 monthly injections of ranibizumab. There were, however, more treatment-related ocular adverse events with abicipar every 8 weeks (17.6%) and abicipar every 12 weeks (22.5%) than with ranibizumab (6.4%). Much of this difference seems attributable to intraocular inflammation, which was reported in 15.1% to 15.7% of eyes receiving abicipar and in 0.3% of eyes receiving ranibizumab. The incidence of this inflammation decreased with time, with 68.7% of intraocular inflammation events occurring after 1 of the first 3 injections. It is important to note, however, that patients experiencing this complication exited the study. Serious ocular and systemic adverse events were evenly distributed across all treatment groups in both studies.⁴³ Improvements were then made to the drug's manufacturing process to reduce the inflammation rate. In the subsequent MAPLE study evaluating the product of this new manufacturing process in treatment-naïve or treatment-experienced eyes with

Table 2. Best-Corrected Visual Acuity Changes From Baseline andProportion of Patients* Maintaining Vision at 52 Weeks in the CEDARand SEQUOIA Phase 3 Trials of Abicipar^{44,45}

| Treatment | n | Mean BCVA Change From Baseline | Patients With Stable Vision [†] , % |
|--|-----|-----------------------------------|---|
| Abicipar 2 mg every 8 weeks | 443 | 8.9 | 96 |
| Abicipar 2 mg every 12 weeks | 442 | 7.4 | 94 |
| Ranibizumab 0.5 mg every 4 weeks | 520 | 9.5 | 97 |

Abbreviation: BCVA, best-corrected visual acuity.

* Finished the full 104 weeks of the study

t Stable vision defined as a loss of < 15 Early Treatment of Diabetic Retinopathy Study letters compared with baseline

nAMD, the overall inflammation rate was 8.9% at 28 weeks, and the rate of severe inflammation was reduced to 1.6%. $^{\rm 46}$

Conbercept

Conbercept is a fusion protein similar to aflibercept, but with an additional VEGF-binding domain. Aflibercept incorporates domain 2 of VEGF receptor 1 and domain 3 of VEGF receptor 2, whereas conbercept adds domain 4 of receptor 2, which decreases the positive charge of the molecule and may reduce adhesion to extracellular matrix.⁴⁷ Conbercept serves as a soluble VEGF receptor decoy, binding to all forms of VEGF as well as placental growth factor, preventing these molecules from reaching their active receptors and thus blocking their activity. It has a similar binding affinity and vitreous half-life to aflibercept.

The phase 3 PHOENIX study was a 12-month, double-masked, sham-controlled trial conducted in 124 subjects in China.⁴⁸ The active arm (n = 81) received 3 monthly injections of conbercept 0.5 mg, followed by quarterly injections thereafter, whereas the sham group (n = 43) received 3 sham injections, followed by quarterly active conbercept injections. The primary end point was mean change in BCVA from baseline to month 3, and was +9.20 letters in the conbercept group and +2.02 letters in the sham group (P < .001). At month 12, after the sham group crossed over to active therapy at month 3, mean BCVA changes were +9.98 and +8.81 letters, respectively (P = .64), demonstrating no significant detriment to the 3-month delay in therapy in the sham group. Most of the reported adverse events were related to the injection process (eg, conjunctival hemorrhage), and increased intraocular pressure occurred in 4.9% of conbercept-treated eyes by month 12. Other clinical trials conducted exclusively in China-where the drug is approved for nAMD treatment–confirm these findings.⁴⁹ Conbercept is currently being evaluated in a pair of 2-year global phase 3 trials-PANDA 1 and PANDA 2-in eyes with treatment-naïve nAMD.^{50,51} The trials compare conbercept 0.5 mg every 8 weeks and 1.0 mg every 12 weeks with aflibercept 2.0 mg every 8 weeks.

Faricimab

Faricimab is a bispecific antibody, in which binding sites for both VEGF-A and angiopoietin-2 (Ang-2) have been incorporated **(Figure 5)**.⁵² Vascular endothelial cells have a tyrosine kinase transmembrane receptor (Tie-2) that can bind the growth factors angiopoietin-1 (Ang-1) and Ang-2.⁵³ Ang-1 binding to the Tie-2 receptor promotes vascular stability and maintains healthy endothelial barrier function to prevent vascular leakage. Ang-2 is elevated in pathophysiologic states and competes with Ang-1 for binding at the Tie-2 receptor. Ang-2 interrupts the vascular





Figure 5. Molecular representation of faricimab⁵²

Abbreviations: Ang-2, angiopoietin-2; Fab, fragment antigen binding; Fc, fragment crystallizable; VEGF-A, vascular endothelial growth factor A. Reprinted from *Ophthalmology*, 126, Sahni J, Patel SS, Dugel PU, et al, Simultaneous inhibition of angiopoietin-2 and vascular endothelial growth factor-A with faricimab in diabetic macular edema: BOULEVARD phase 2 randomized trial, 1155-1170, Copyright 2019, with permission from Elsevier.

stabilization functions of Ang-1, leading to breakdown of the bloodretinal barrier and intracellular inflammation.⁵⁴ Thus, by binding and blocking the effects of both VEGF and Ang-2, faricimab has a dual mechanism of action that may be beneficial in nAMD.

Faricimab has completed phase 2 clinical development for nAMD in a pair of studies: AVENUE and STAIRWAY. 55,56 STAIRWAY enrolled 76 patients with treatment-naïve nAMD and randomly assigned them to receive either faricimab 6.0 mg every 12 or 16 weeks or ranibizumab 0.5 mg every 4 weeks for 12 months.⁵⁷ All faricimab patients received 4 monthly loading doses, with sham injections given as needed to preserve double masking. At week 24, 12 weeks after the last loading dose, 65% of eyes receiving either dose of faricimab exhibited no disease activity, as assessed using prespecified criteria.⁵⁷ The mean changes from baseline in BCVA at month 12 were +10.08, +11.42, and +9.59 letters with faricimab every 12 weeks, faricimab every 16 weeks, and ranibizumab, respectively, and the mean changes in OCT central subfield thickness from baseline to month 12 were -138.5, -122.5, and -129.9 µm, respectively. No serious ocular adverse events occurred in any eyes, and nonserious ocular adverse events occurred with similar frequency in all 3 groups.⁵⁷ Phase 3 trials, **TENAYA** and LUCERNE, are currently proceeding and are fully enrolled.58,59

Port Delivery System

The port delivery system (PDS) is a refillable implant inserted through the pars plana into the vitreous cavity that delivers a customized formulation of ranibizumab in slow-release fashion into the vitreous for extended periods of time.⁶⁰ In the phase 2 LADDER trial (N = 220), PDS filled with ranibizumab 100 mg/mL provided nAMD disease control for a median of 15 months before the first refill and produced mean BCVA gains from baseline to month 9 of +5.0 letters, which was comparable to that seen with monthly ranibizumab 0.5 mg (+3.9 letters); note that these patients were previously treated with anti-VEGF therapy to confirm responsiveness before entering this trial, accounting for the smaller VA gains than would be expected in treatment-naïve eyes.⁶⁰ Changes in OCT central foveal thickness to month 9 were also similar in the PDS 100 mg/mL and monthly ranibizumab injection groups. Vitreous hemorrhage occurred in approximately 50% of eyes before optimization of the surgical procedure midway

through the phase 2 program, reducing this rate to approximately 4.5%. The phase 3 **ARCHWAY** trial is currently under way; data are expected this year.⁶¹

Other Therapies

Other promising therapies, including both drugs and sustained drug delivery systems, are in earlier stages of development. KSI-301 is an antibody biopolymer conjugate, a 950-kDa protein with an ocular concentration at 3 months after dosing approximately 1000-fold higher than that of aflibercept, which is designed to block all isoforms of VEGF-A.⁶²

GB-102, a depot formulation of sunitinib, a multiple receptor tyrosine kinase inhibitor, acts as a potent pan-VEGF (VEGF-A, -B, -C, and -D) inhibitor. Sunitinib is currently approved in an oral form as a chemotherapy agent for solid tumors.⁶³ Delivered to the vitreous cavity via a 27G needle, the depot slowly releases therapeutic drug levels for up to 6 months between doses, effectively blocking all 3 VEGF receptors (VEGFR-1, -2, and -3).⁶³

Gene Therapy

Several innovative gene therapies for eyes with nAMD are also in development. RGX-314 is a delivery system that uses an adenoassociated virus 8 to deliver a gene encoding an anti-VEGF antibody fragment similar to ranibizumab.⁶⁴ Following either subretinal or suprachoroidal injection, the anti-VEGF antibody fragment can be detected 2 and 7 weeks postinjection in both retina and retinal pigment epithelium/choroid in rat eyes, with a dose-response relationship demonstrating higher tissue levels after 2 injections vs 1 injection. Another gene therapy approach is using an in-office, intravitreal ADVM-022, an AAV.7m8 vector designed to deliver gene encoding for aflibercept.⁶⁵ This therapy has been shown to produce a vitreous aflibercept concentration consistent with therapeutic levels in humans and to prevent CNV in nonhuman primates pretreated 13 months before the experimental induction of CNV.

PANEL DISCUSSION: LOOKING TO THE FUTURE OF THERAPIES FOR nAMD

Dr Singer: In your opinion, does using a novel molecule to target VEGF in the eye translate to better efficacy and treatment longevity?

Dr Khanani: When evaluating a molecule, 3 key attributes come to mind: molar dosing, binding affinity, and pharmacokinetics in the eye. We do not know which of these factors matters the most, but I think that a combination of all 3 should lead to better efficacy and treatment longevity in the clinic.

Dr Eichenbaum: I think that the "proof is in the pudding", so to speak. These novel designs are innovative and interesting, but the clinical data are most informative, in my opinion.

Dr Holekamp: I agree with Dr Eichenbaum. It will be interesting to see what happens with some of the more novel designs that are not just slightly modified anti-VEGF monoclonal antibodies. The number of patients who can be maintained at a dosing interval longer than every 8 weeks is an important piece of data to consider because it may allow patients to follow up less frequently when using a new agent, thus decreasing the overall burden. Clinical trials have shown differences among new and emerging agents in this regard.

Dr Khanani: As far as data are concerned, resolution of fluid is the most important to me because that is what I treat in the clinic. That being said, in my opinion, agents that dry the fluid better will also be more durable.

Dr Holekamp: Improved fluid resolution does not always correlate with improved VA,⁶⁶ which is arguably more relevant to the patient.



Dr Singer: That is true. We are still trying to figure out as a community the relevance of different types of retinal fluid.

Dr Khanani: Safety is also important. As we learn more about the inflammation associated with new and emerging therapies, we might be able to identify patients who are more likely to experience inflammation. In the end, we have to balance safety and efficacy of any treatment option we provide to our patients.

CASE 1: FLUID RECURRENCE AFTER EXTENSION

From the Files of Michael Singer, MD

A 71-year-old male presented with acute loss of VA. His past medical history included type 2 diabetes and systemic hypertension, and no prior ocular history. On examination, his VA was 20/20 OD and 20/150 OS. He had 3+ nuclear sclerosis of both lenses. He had drusen in both maculae, and retinal pigment epithelium changes and fluid OS (Figure 6A). The OCT image confirmed subretinal fluid OS (Figure 6B).

Three monthly injections of aflibercept were given, resulting in macular drying **(Figure 7A)** and VA improvement to 20/40. He was extended to 8 weeks and seen at 20 weeks, at which time the fluid recurred and VA worsened slightly to 20/50 **(Figure 7B)**.



Figure 6. Color fundus photograph (A) and optical coherence tomography image (B) of the left eye of the patient presented in Case 1, demonstrating subretinal fluid



Figure 7. Optical coherence tomography images of the left eye of the patient in Case 1 at 12 weeks (A) and 20 weeks (B) after an extension to 8 weeks, showing recurrence of subretinal fluid

Discussion

Dr Singer: This patient failed an extension to 2 months after 3 monthly aflibercept injections. What would you do now?

Dr Holekamp: Although I can tolerate a little bit of subretinal fluid, vision here matters. Because vision declined at the same time that subretinal fluid worsened, I think it is time to shorten the treatment interval for this patient and try to get the macula dry. I would do this regardless of anti-VEGF agent being used.

Dr Eichenbaum: Would you consider tolerating that fluid and leaving his macula like this? His VA is still surprisingly good with that amount of subretinal fluid. One might opt to tolerate the fluid, given the relative Snellen VA stability, but I would prefer the macula be drier than that because the patient was anatomically dryer with better VA at the preceding visit.

Dr Singer: I agree. Given that he would need ongoing monthly injections with aflibercept, I opted to switch to brolucizumab in

hopes of achieving and maintaining a dry macula and improved VA with fewer injections. We have accomplished this with brolucizumab extended to every 10 weeks now.

Case 1 Summary

Dr Singer: Our treatment goal is to eliminate as much fluid as possible. Residual intraretinal fluid is associated with loss of vision. I believe that the subretinal fluid shadow seen on OCT images may consist of different factors early in the disease process as compared with later in disease process. Although the goal is still to get the retina as dry as possible, I am willing to settle for residual subretinal fluid in the longer term so long as vision does not worsen.

CASE 2: PERSISTENT FLUID

From the Files of Arshad M. Khanani, MD, MA

A 70-year-old female has been under care for nAMD for 7 years. She had received a total of 15 bevacizumab injections, 8 ranibizumab injections, and 37 aflibercept injections, 8 of which were given in 2019. In this patient, VA fluctuates between 20/25 and 20/40. Despite multiple attempts to extend the interval between injections, she required injections consistently every 4 to 5 weeks. **Figure 8** shows an OCT image of her eye 6 weeks after an aflibercept injection, revealing persistent fluid when treatment was extended beyond 5 weeks.



Figure 8. Optical coherence tomography image of the patient presented in Case 2, showing persistent fluid when aflibercept therapy was extended to 6 weeks

Discussion

Dr Khanani: Given the tremendous treatment burden–visits and injections every 5 weeks, even with a longer-acting anti-VEGF drug such as aflibercept–what would you do now?

Dr Holekamp: It is favorable that this is subretinal fluid because even recurrent subretinal fluid can be compatible with good vision long term in patients. We know this from the FLUID study.⁶⁷ This is, however, a significant amount of subretinal fluid. I would make a change and switch agents.

Dr Eichenbaum: Because you have already tried every other anti-VEGF agent for a minimum of 8 injections, I would discuss a switch to brolucizumab with the patient in hopes of extending beyond 5 weeks.

Dr Khanani: The patient and I agreed that brolucizumab was the best next step. Four weeks after the last affibercept injection, BCVA was 20/30 and the OCT image revealed trace subretinal fluid and a small pigment epithelial detachment **(Figure 9A)**. Brolucizumab was injected at that time, and at 6 weeks postinjection, BCVA was 20/25 and the macula was completely dry **(Figure 9B)**. We reassessed every 2 weeks to determine the optimal extension interval, and the VA and OCT remained stable until week 14 **(Figure 9C)**, when the appearance of subtle subretinal fluid (and BCVA 20/30) led us to re-treat. With brolucizumab, she can now be injected every 3 months instead of every 5 weeks with aflibercept.

Dr Singer: Not only is her disease better, but her treatment burden is significantly reduced. Longer-acting anti-VEGF therapies such as this one are very likely to improve overall adherence to follow-up and therapy, especially among our more rural patients, who live far from our treatment centers.



Figure 9. Optical coherence tomography image sequence of the patient presented in Case 2: (A) before brolucizumab; (B) 6 weeks after brolucizumab; and (C) 14 weeks after brolucizumab

Case 2 Summary

Dr Khanani: This case highlights the treatment burden and frequent visits required by some patients with nAMD. As with this patient, there are patients who require monthly anti-VEGF injections, or else their disease is not controlled. Emerging molecules may be beneficial in addressing this unmet need, as shown by an excellent response to recently approved brolucizumab.¹²

CASE 3: SWITCHING ANTI-VEGF AGENTS

From the Files of David Eichenbaum, MD

An otherwise healthy 67-year-old male presented with distortion and blurry vision OS. His BCVA was 20/25 OD and 20/63 OS. **Figure 10A** shows his baseline OCT images and reveals subretinal fluid. Three monthly loading doses of aflibercept were given, after which TAE was initiated. Extension was attempted 3 times over 2 years, but the patient never achieved an interval beyond 5 weeks because of recurrent subretinal fluid with stable BCVA of 20/50 after 6 weeks (**Figure 10B**). The patient was unhappy with the frequency of visits and injections and was switched to brolucizumab. Five weeks later, BCVA was 20/32 and there was no fluid (**Figure 10C**), so the patient was re-treated and extended to 6 to 7 weeks. At that visit—the most recent encounter—BCVA and the OCT images remained stable (**Figure 10D**), so the patient was re-treated and extended to 7 to 8 weeks.

Discussion

Dr Eichenbaum: What are your indications for switching to a different anti-VEGF agent?

Dr Singer: When initiating therapy in treatment-naïve patients, I give a minimum of 4 injections, and if macular dryness is not achieved, I consider switching to a different agent. Also, as this case illustrates, even if we do achieve macular dryness, I would consider switching from a shorter-acting agent to a longer-acting agent to decrease the treatment burden.

Dr Holekamp: We know from all the randomized clinical trials that even with frequent, consistent anti-VEGF treatments, some eyes never achieve dryness. Thus, it is exciting to see the development of new agents that may allow us to reach 2 goals: better drying and decreased burden. These are the 2 reasons I would consider switching to a new agent.

Dr Khanani: I usually consider switching after 3 to 6 monthly injections if I see persistent disease activity. This patient has clearly benefited from switching to brolucizumab. In the future, we will likely have other options to switch to—such as abicipar,



Figure 10. Optical coherence tomography image sequence of the left eye of the patient presented in Case 3: (A) at the time of nAMD diagnosis; (B) at the time of first brolucizumab injection; (C) 5 weeks after beginning brolucizumab; and (D) 11 weeks after beginning brolucizumab

faricimab, and PDS—and patients such as the one in Case 3 may benefit from these options. Each agent or delivery system may have a different efficacy and safety profile, and we always have to balance safety and efficacy whenever we use a new agent.

Dr Eichenbaum: How rapidly do you think physicians and patients will embrace some of the emerging therapies that may offer even longer durations of action than those of our current drugs?

Dr Singer: In my experience, patients will do whatever it takes to preserve their VA. It is exciting to have so many current and emerging therapies with such a variety of attributes. We are approaching the era of individualized therapy for nAMD, in which we will have the luxury of many treatments from which to choose and can select the agent that best fits the needs of a particular patient.

Case 3 Summary

Dr Eichenbaum: Switching agents in patients who are intolerant or nonresponsive to frequent injections is a reasonable strategy to try to mitigate the burden of nAMD care. As we see more, potentially longer-lasting agents come into the treatment space, we can hope to have more options with a lower treatment burden for patients.

TAKE-HOME POINTS

- The treatment burden in nAMD limits optimal anti-VEGF dosing rates in real-world clinical practice, resulting in suboptimal VA outcomes compared with those reported in clinical trials
- Multiple studies now demonstrate that the TAE dosing regimen for anti-VEGF therapy provides similar VA outcomes as regular injections every month or 3 months, with a significantly lower injection rate over time
- Novel drugs, devices, and platforms with longer durations of action attributable to greater tissue penetration and/or higher binding affinities vs current therapies provide options for optimizing VA outcomes with a lesser treatment burden
- New drugs and novel delivery systems provide functional (VA), structural (OCT), and safety outcomes comparable to those achieved with anti-VEGF injections administered every 4 to 12 weeks



REFERENCES

- Wong WL, Su X, Li X, et al. Global prevalence of age-related macular degeneration and disease burden projection for 2020 and 2040: a systematic review and meta-analysis. Lancet Glob Health. 2014;2(2):e106-e116.
- Lim LS, Mitchell P, Seddon JM, Holz FG, Wong TY. Age-related macular degeneration. Lancet. 2012;379(9827):1728-1738. 2
- Ferris FL 3rd, Fine SL, Hyman L. Age-related macular degeneration and blindness due to neovascular maculopathy. Arch Ophthalmol. 1984;102(11):1640-1642.
- Novartis AG. Novartis provides update on use and safety of Beovu (brolucizumab). Accessed May 18, 2020. https://www.brolucizumab.info/ 4.
- Holz FG, Tadayoni R, Beatty S, et al. Key drivers of visual acuity gains in neovascular age-related macular degeneration in real life: findings from the AURA study. Br J Ophthalmol. 2016;100(12):1623-1628.
- Rosenfeld PJ, Brown DM, Heier JS, et al; MARINA Study Group. Ranibizumab for neovascular
- Kosenteid PJ, Brown DM, Heler JS, et al; MAKINA Study Croupt. KaniDizuma for neovascular age-related macular degeneration. *N Engl J Med*. 2006;355(14):1419-1431.
 Brown DM, Michels M, Kaiser PK, Heier JS, Sy JP, lanchulev T; ANCHOR Study Group. Ranibizumab versus verteporfin photodynamic therapy for neovascular age-related macular degeneration: two-year results of the ANCHOR study. *Ophthalmology*. 2009;116(1):57-65.e5.
 Martin DF, Maguire MG, Fine SL, et al; Comparison of Age-Related Macular Degeneration Treatments Trials (CATT) Research Group. Ranibizumab and bevacizumab for treatment of
- neovascular age-related macular degeneration: two-year results. Ophthalmology 2012:119(7):1388-1398.
- Sharma S, Toth CA, Daniel E, et al; Comparison of Age-Related Macular Degeneration Treatments Trials Research Group. Macular morphology and visual acuity in the second year of the Comparison of Age-Related Macular Degeneration Treatments Trials. *Ophthalmology*. 2016:123(4):865-875.
- Heier JS, Brown DM, Chong V, et al; VIEW 1 and VIEW 2 Study Groups. Intravitreal aflibercept (VEGF trap-eye) in wet age-related macular degeneration. Ophthalmology. 2012;119(12): 2537-2548.
- Jaffe GJ, Kaiser PK, Thompson D, et al. Differential response to anti-VEGF regimens in age-related macular degeneration patients with early persistent retinal fluid. *Ophthalmology*. 11. 2016-123(9)-1856-1864
- Dugel PU, Koh A, Ogura Y, et al; HAWK and HARRIER Study Investigators. HAWK and 12. HARRIER: phase 3, multicenter, randomized, double-masked trials of brolucizumab for neovascular age-related macular degeneration. Ophthalmology. 2020;127(1):72-84.
- Beovu. Package insert. Novartis Pharmaceuticals Corporation; 2020. The Medical XChange. American Academy of Ophthalmology (AAO) 2018. Accessed May 8, 2020. https://themedicalxchange.com/en/2018/11/18/aao_2390_for-macular-degeneration-new-data-validates-less-frequent-anti-vegf-dosing/#
- Dugel PU, et al. Phase 3, randomized, double-masked, multi-center trials of brolucizumab versus aflibercept for neovascular AMD: 96-week results from the HAWK and HARRIER 15. studies. Paper presented at: 2018 Annual Meeting of the American Academy of Ophthalmology; October 27-30, 2018; Chicago, IL.
- Ho AC, Busbee BG, Regillo CD, et al; HARBOR Study Group. Twenty-four-month efficacy and safety of 0.5 mg or 2.0 mg ranibizumab in patients with subfoveal neovascular age-related macular degeneration. *Ophthalmology*. 2014;12(11):2181-2192. 16.
- 17. ninisizuma vs. affiliercept for neovascular age-related macular degeneration: data from an observational study. *Ophthalmology*. 2016;123(12):2545-2553. Holz FG, Bandello F, Gillies M, et al; LUMINOUS Steering Committee. Safety of ranibizumab in routine clinical practice: 1-year retrospective pooled analysis of four European neovascular
- 18.
- AMD registries within the LUMINOUS programme. *Br J Ophthalmol.* 2013;97(9):1161-1167. Holz FG, Tadayoni R, Beatty S, et al. Multi-country real-life experience of anti-vascular 19 endothelial growth factor therapy for wet age-related macular degeneration. Br J Ophthalmol. 2015;99(2):220-226.
- 20 Writing Committee for the UK Age-Related Macular Degeneration EMR Users Group
- The neovascular age-related macular begeneration EMR Users Broup. The neovascular age-related macular degeneration database: multicenter study of 92 976 ranibizumab injections: report 1: visual acuity. *Ophthalmology*. 2014;121(5):1092-1101. Kim LN, Mehta H, Barthelmes D, Nguyen V, Gillies MC. Metaanalysis of real-world outcomes of intravitreal ranibizumab for the treatment of neovascular age-related macular degeneration. *Retina*. 2016;36(8):1418-14431. 21.
- Okada M, Kandasamy R, Chong EW, McGuiness M, Guymer RH. The treat-and-extend 22 injection regimen versus alternate dosing strategies in age-related macular degeneration: a systematic review and meta-analysis. *Am J Ophthalmol.* 2018;192:184-197. Holekamp NM, Liu Y, Yeh WS, et al. Clinical utilization of anti-VEGF agents and disease
- 23 monitoring in neovascular age-related macular degeneration. Am J Ophthalmol. 2014;157(4): 825-833.e1.
- Singer MA, Awh CC, Sadda S, et al. HORIZON: an open-label extension trial of ranibizumab for choroidal neovascularization secondary to age-related macular degeneration Ophthalmology. 2012;119(6):1175-1183.
- Martin DF, Maguire MG, Ying G-S, Grunwald JE, Fine SL, Jaffe GL; CATT Research Group. Ranibizumab and bevacizumab for neovascular age-related macular degeneration.
- N Engl J Med. 2011;364(20):1897-1908. Ying G-S, Huang J, Maguire MG, et al; Comparison of Age-Related Macular Degeneration Treatments Trials Research Group. Baseline predictors for one-year visual outcomes with 26. ranibizumab or bevacizumab for neovascular age-related macular degeneration.
- Ophthalmology. 2013;120(1):122-129. Ying G-S, Maguire MG, Daniel E, et al; Comparison of Age-Related Macular Degeneration Treatments Trials (CATT) Research Group. Association of baseline characteristics and early vision response with 2-year vision outcomes in the Comparison of AMD Treatments Trials
- (CATT) Ophthalmology. 2015;122(12):2523-2531.e1.
 28. Kim JE. Performance and usability of a self-operated home opctical coherence tomography (OCT) system: NOTAL-OCT V2.5 (NOV2.5). ASRS MultiMedia Library. 2019. Accessed May 8, 2020. https://multimedia.asrs.org/media/performance-and-usability-of-a-self-operatedhome--7158
- Lucentis. Package insert. Genentech, Inc; 2019
- Eylea. Package insert. Regeneron Pharmaceuticals, Inc; 2019. Schmucker CM, Rücker G, Sommer H, et al. Treatment as required versus regular monthly 30 treatment in the management of neovascular age-related macular degeneration: a systematic review and meta-analysis. *PLoS One*. 2015;10(9):e0137866.
- Wykoff CC, Ou WC, Brown DM, et al. TREX-AMD Study Group. Randomized trial of treat-and-extend versus monthly dosing for neovascular age-related macular degeneration: 2-year results of the TREX-AMD study. *Ophthalmol Retina*, 2017;1(4):314-321. Silva R, Berta A, Larsen M, Macfadden W, Feller C, Monés J; TREND Study Group. Treat-and-32
- 33. extend versus monthly regimen in neovascular, nentes 2, mones 2, more adapted and the second second
- 34 the randomized Canadian Treat-and-Extend Analysis Trial With Ranibizumab study. Ophthalmology. 2019;126(6):841-848.
- 35. Berg K, Hadzalic E, Gjertsen I, et al. Ranibizumab or bevacizumab for neovascular age related macular degeneration according to the Lucentis Compared to Avastin study treat-and-extend protocol: two-year results. *Ophthalmology*. 2016;123(1):51-59.

- 36. DeCroos FC, Reed D, Adam MK, et al. Treat-and-extend therapy using affibercept for neovascular age-related macular degeneration: a prospective clinical trial. Am J Ophthalmol. 2017;180:142-150.
- Ohji M, Takahashi K, Okada AA, Kobayashi M, Matsuda Y, Terano Y; ALTAIR Investigators. Efficacy and safety of intravitreal aflibercept treat-and-extend regimens in exudative age-related macular degeneration: 52- and 96-week findings from ALTAIR : a randomized controlled trial. Adv Ther. 2020;37(3):1173-1187.
- Stumpp MT, Binz HK, Amstutz P. DARPins: a new generation of protein therapeutics Drug Discov Today. 2008;13(15-16):695-701. 38
- Wetzel SK, Settanni G, Kenig M, Binz HK, Plückthun A. Folding and unfolding mechanism of highly stable full-consensus ankyrin repeat proteins. J Mol Biol. 2008;376(1):241-257.
- A. Rodrigues GA, Mason M, Christie L-A, et al. Functional characterization of abicipar-pegol, an anti-VEGF DARPin therapeutic that potently inhibits angiogenesis and vascular permeability. *Invest Ophthalmol Vis Sci.* 2018;9(15):5836-5846.
 Sharma A, Kumar N, Kuppermann BD, Bandello F. Abicipar pegol: the non-monoclonal
- Shariha A, Kuhari N, Kuperinaini DD, Darideni P, Kolcipa Degol: the normalization of the state o
- 43. Molecular Partners. Allergan and Molecular Partners present late-breaking data from phase Studies of investigational abicipar pegol in neovascular wet age-related macular degeneration. Press release. October 11, 2019. Accessed May 8, 2020. https://www.molecularpartners.com/allergan-and-molecular-partners-present-late-breaking-data-from-phase-3-studies-of-investigational-abicipar-pegol-in-neovascular-
- wet-age-related-macular-degeneration/ 44. Kunimoto D, Yoon YH, Wykoff CC, et al; CEDAR and SEQUOIA Study Groups. Efficacy and safety of abicipar in neovascular age-related macular degeneration: 52-week results of phase 3 randomized controlled study. Accepted manuscript. Published online March 30,
- 2020. Ophthalmology. doi:10.1016/j.ophtha.2020.03.035
 Khurana R. Abicipar for neovascular age-related macular degeneration: two-year results from CEDAR and SEQUOIA phase 3 clinical trials. Paper presented at: 2019 Annual Meeting of the American Academy of Ophthalmology; October 12-15, 2019; San Francisco, CA.
- PRNewswire. Allergan and Molecular Partners announce topline safety results from MAPLE study of abicipar pegol. News release. April 2, 2019. Accessed May 8, 2020. https://www.prnewswire.com/news-releases/allergan-and-molecular-partners-announce-topline-safety-results-from-maple-study-of-abicipar-pegol-300822353.html
- Lu X, Sun X. Profile of conbercept in the treatment of neovascular age-related macular degeneration. *Drug Des Devel Ther.* 2015;9:2311-2320.
 Liu K, Song Y, Xu G, et al; PHOENIX Study Group. Conbercept for treatment of neovascular age-related macular degeneration: results of the randomized phase 3 PHOENIX study. Am J Ophthalmol. 2019;197:156-167. Zhang J, Liang Y, Xie J, et al. Conbercept for patients with age-related macular
- degeneration: a systematic review. *BMC Ophthalmol.* 2018;18(1):142. 50. Chengdu Kanghong Biotech Co, Ltd. Efficacy and safety trial of conbercept intravitreal
- injection for neovascular age-related macular degeneration (PANDA-1). ClinicalTrials.gov. July 5, 2018. Updated October 10, 2018. Accessed May 8, 2020. https://clinicaltrials.gov/ ct2/show/NCT03577899
- Chengdu Kanghong Biotech Co, Ltd. Efficacy and safety trial of conbercept intravitreal 51. injection for neovascular age-related macular degeneration (PANDA-2). ClinicalTrials.gov. August 15, 2018. Updated March 5, 2019. Accessed May 8, 2020. https://clinicaltrials.gov/ start (J Los Salar), and start (J Los Salar),
- 52. endothelia growth factor-A with faricimab in diabetic more angropoledin-2 and vascular endothelial growth factor-A with faricimab in diabetic more angropoledin-2 and vascular andomized trial. *Ophthalmology*. 2019;126(8):1155–1170. Sato TN, Tozawa Y, Deutsch U, et al. Distinct roles of the receptor tyrosine kinases Tie-1 and Tie-2 in blood vessel formation. *Nature*. 1995;376(6535):70–74.
- 53.
- Saharinen P, Eklund L, Alitalo K. Therapeutic targeting of the angiopoietin–TIE pathway. Nat Rev Drug Discov. 2017;16(9):635-661. 54.
- Not Rev Drug Discov. 2017;16(9):653–661.
 St. Hoffman-La Roche. A proof-of-concept study of faricimab (R06867461) in participants with choroidal neovascularization (CNV) secondary to age-related macular degeneration (AMD) (AVENUE). ClinicalTrials.gov/ct2/show/NCT02484690
 Hoffmann-La Roche. Study to evaluate faricimab (R06867461: RG7716) for extended durability in the tractment of acoustoput reas provided the moving degeneration (AMD)
- Administricta Roche. Study to evaluate faitchinal (Robos/46); RO7716) for extended durability in the treatment of neovascular age related macular degeneration (nAMD) (STAIRWAY). ClinicalTrials.gov. February 1, 2017. Updated January 14, 2020. Accessed May 8, 2020. https://clinicaltrials.gov/ct2/show/NCT03038880 Khanani AM. Simultaneous inhibition of VEGF and Ang-2 with faricimab in neovascular AMD: STAIRWAY phase 2 results. Paper presented at: 2018 Annual Meeting of the American Academy of Opthalmology: October 27-30, 2018; Chicago, IL.
- Hoffmann-La Roche. A study to evaluate the efficacy and safety of faricimab in participants with neovascular age-related macular degeneration (TENAYA). ClinicalTrials.gov. January 30, 2019. Updated February 25, 2020. Accessed May 8, 2020. https://clinicaltrials.gov/ct2/show/ NCT03823287
- Hoffmann-La Roche. A study to evaluate the efficacy and safety of faricimab in participants with neovascular age-related macular degeneration (LUCERNE). ClinicalTrials.gov. January 30, 2019. Updated April 21, 2020. Accessed May 8, 2020. https://clinicaltrials.gov/ct2/show/ NCT03823300
- Campochiaro PA, Marcus DM, Awh CC, et al. The port delivery system with ranibizumab for neovascular age-related macular degeneration: results from the randomized phase 2 Ladder clinical trial. *Ophthalmology*. 2019;126(8):141–1154. Hoffmann-La Roche. A phase III study to evaluate the port delivery system implant with 60
- 61. ranibizumab compared with monthly ranibizumab injections in participants with wet age related macular degeneration (Archway). ClinicalTrials.gov. September 19, 2018. Updated
- March 3, 2020. Accessed May 8, 2020. https://clinicaltrials.gov/ct2/show/NCT03677934 Securities and Exchange Commission. Form S-1 Registration Statement: Kodiak Sciences Inc. Accessed May 8, 2020. https://www.sec.gov/Archives/edgar/data/1468748/ 62 000119312518269221/d516071ds1.htm
- Roach L. Innovative drugs in the pipeline: a brief overview. EyeNet. February 2016. Accessed May 8, 2020. https://www.aao.org/eyenet/article/innovative-drugs-in-pipeline-brief-
- overview?february-2016 64. Ding K, Shen J, Hafiz Z, et al. AAV8-vectored suprachoroidal gene transfer produces widespread ocular transgene expression. *J Clin Invest.* 2019;130(11):4901-4911. Grishanin R, Vuillemenot B, Sharma P, et al. Preclinical evaluation of ADVM-022, a novel gene
- 65. therapy approach to treating wet age-related macular degeneration. Mol Ther. 2019;27(1): 118-129.
- Regillo CD, Busbee BG, Ho AC, Ding B, Haskova Z. Baseline predictors of 12-month treatment response to ranibizumab in patients with wet age-related macular degeneration. Am J Ophthalmol. 2015;160(5):1014-1023.e2. Guymer RH, Markey CM, McAllister IL, Gillies MC, Hunyor AP, Arnold JJ; FLUID Investigators.
- 67 Tolerating subretinal fluid in neovascular age-related macular degeneration treated with ranibizumab using a treat-and-extend regimen: FLUID study 24-month results. Ophthalmology. 2019;126(5):723-734.

INSTANT CME CERTIFICATE AVAILABLE WITH ONLINE TESTING AND COURSE EVALUATION AT

HTTPS://TINYURL.COM/SAVINGSIGHTNAMD



CME POST TEST QUESTIONS

To obtain AMA PRA Category 1 Credit[™] for this activity, complete the CME Post Test and course evaluation **online** at **https://tinyurl.com/savingsightnAMD**. (Paper submissions cannot be processed.) Upon successful completion of the post test and evaluation, you will be able to generate an instant certificate of credit.

See detailed instructions at To Obtain AMA PRA Category 1 Credit™ on page 2.

- 1. Which drug was shown to maintain BCVA better than both sham injection and PDT with verteporfin in phase 3 trials?
 - a. Bevacizumab
 - b. Ranibizumab
 - c. Aflibercept
 - d. Brolucizumab
- 2. Aflibercept dosed every ______ demonstrated noninferior maintenance of BCVA compared with monthly ranibizumab therapy.
 - a. Month
 - b. 2 months
 - c. 3 months
 - d. Both a and b
- 3. In the phase 3 HAWK and HARRIER trials,
- of eyes were maintained on brolucizumab 6 mg every 12 weeks at month 12.
 - a. 12% to 16%
 - b. 34% to 37%
 - c. 51% to 56%
 - d. 75% to 78%
- 4. In the CEDAR and SEQUOIA phase 3 trials comparing abicipar dosed every 2 or 3 months with monthly ranibizumab, the 24-month BCVA and OCT outcomes were comparable in all 3 groups. This was achieved with ____ quarterly injections of abicipar and ____ monthly injections of ranibizumab.
 - a. 5, 14
 - b. 25, 10
 - c. 10, 25
 - d. 16, 24
- 5. Which of the following characteristics likely contributes to abicipar's extended duration of action (up to 12 weeks)?
 - a. High target specificity from its ankyrin repeat domain
 - b. Higher molar dosing due to its small size
 - c. Long half-life in ocular tissues
 - d. All the above
- 6. Faricimab was shown in the phase 2 STAIRWAY trial to suppress nAMD disease activity for up to 16 weeks following each dose.
 - One possible explanation for this extended duration of activity is: a. Faricimab is a small DARPin with a relatively long ocular half-life
 - b. Faricimab inhibits the activity of VEGF-A, VEGF-B, and VEGF-C
 - c. Faricimab inhibits both the VEGF and Tie-2 pathways
 - d. Faricimab inhibits platelet-derived growth factor

- 7. The PDS is a sustained-release platform with a median time to refill of up to every _____ months when delivering ranibizumab 100 mg/mL.
 - a. 3
 - b. 6
 - c. 10
 - d. 15
- 8. Which of the following anti-VEGF dosing strategies typically has the least favorable VA outcomes?
 - a. Regular injections every 1 to 3 months as indicated for specific drugs
 - b. TAE to increase the interval between injections while still suppressing disease activity
 - c. As needed when VA or OCT image appearance worsens
 - d. Every 10 weeks regardless of OCT image appearance
- 9. A patient with newly diagnosed nAMD has no disease activity after 3 monthly loading doses of anti-VEGF therapy. According to the TAE dosing strategy, the next dose should be given _____ weeks later.
 - a. 2
 - b. 4
 - c. 6
 - d. 9
- 10. A patient's nAMD was stable with extension of anti-VEGF injections to every 10 weeks, but then recurrent subretinal fluid was seen at a scheduled 10-week follow-up. What is the best next step?
 - a. Give 3 additional monthly loading doses and then begin extending the treatment interval
 - b. Re-treat and reevaluate in 4 weeks
 - c. Reevaluate in 2 weeks without re-treating
 - d. Re-treat and reevaluate in 8 weeks